Author’s response to reviews

Title: Pathologic fracture of the distal radius in a 25 year old patient with a large unicameral bone cyst.

Authors:

Felix Massen (felix.massen@med.uni-muenchen.de)
Sebastian Baumbach (sebastian.baumbach@med.uni-muenchen.de)
Elias Volkmer (elias.volkmer@med.uni-muenchen.de)
Wolf Mutschler (Wolf.Mutschler@med.uni-muenchen.de)
Stefan Grote (Stefan.Grote@med.uni-muenchen.de)

Version: 2 Date: 31 January 2014

Author’s response to reviews:

In this first revision we changed/added the following parts to the Manuscript according:

1. Included a title page with all authors and addresses
2. Restructured the abstract with separate sections: background, case presentation and conclusion
3. Included the 'authors contributions' section
4. Included a 'Consent' section with the patients consent